

Clinically suspect arthralgia: unraveling the development of rheumatoid arthritis

Wouters, F.

Citation

Wouters, F. (2023, February 21). *Clinically suspect arthralgia: unraveling the development of rheumatoid arthritis*. Retrieved from https://hdl.handle.net/1887/3564199

Version:	Publisher's Version
License:	<u>Licence agreement concerning inclusion of doctoral</u> <u>thesis in the Institutional Repository of the University</u> <u>of Leiden</u>
Downloaded from:	https://hdl.handle.net/1887/3564199

Note: To cite this publication please use the final published version (if applicable).





Summary and discussion

In this thesis two main aims were addressed. It has long been established that early treatment of rheumatoid arthritis (RA) improves disease outcomes. In Part I of this thesis we therefore further investigated the early detection of at-risk individuals by studying a large cohort of patients with clinically suspect arthralgia (CSA). We explored the value of two easy clinical tests, their potential to detect underlying inflammatory processes and to predict disease progression. In addition we investigated the presence of subclinical synovitis on imaging as starting point for treatment with disease modifying anti-rheumatic drugs (DMARDs) and the value of magnetic resonance imaging (MRI) detected erosions as new predictor for RA-development. In **Part II** of this thesis we aimed to determine which disease processes are involved in the different phases of RA-development. Knowledge on disease pathogenesis and timing of influencing factors can help to better target treatment during RA-development. We therefore evaluated whether autoantibody-response maturation occurred during the phase of CSA, and investigated the timing of genetic risk factor human leukocyte antigen-shared epitope (HLA-SE) and environmental risk factor smoking during the development of autoantibody-positive disease.

Part I – Prediction and early detection of rheumatoid arthritis

Summary Chapter 2 and 3

The value of imaging in the prediction of RA has often been investigated. Subclinical inflammation can be detected even before the occurrence of clinically detectable arthritis and has been shown to predict disease progression.^{1,2} However, imaging modalities as ultrasonography (US) and MRI are costly, time consuming and not always available. Moreover, it was hypothesized that subclinical features might underly clinical manifestations that are considered risk factors for development of RA. In **Chapter 2** we therefore investigated difficulties making a fist, one of the factors comprising the EULAR definition of arthralgia suspicious for progression to RA.³ Although fist problems are considered a risk factor for RA-development in patients presenting with CSA, its predictive value and underlying cause were unclear. Difficulties making a fist was assessed in two ways: 1) fist closure was evaluated by visual inspection of the ability to completely close the fist, all fingertips touching the palm, 2) fist strength was determined by the patient squeezing the assessor's fingers. Incomplete fist closure and a decreased fist strength were both independently associated with progression to clinically apparent inflammatory arthritis (IA), though incomplete fist closure had a higher predictive value and better reliability. Fist problems associated significantly with flexor tenosynovitis; incomplete fist closure associated predominantly with flexor tenosynovitis of metacarpophalangeal (MCP) joints, whereas decreased fist strength more strongly related to flexor tenosynovitis of the wrist. These results indicate that difficulties making a fist, and predominantly fist closure, is easily assessable in clinical practice and can provide information on both risk assessment for disease progression as well as underlying flexor tenosynovitis.

In addition to fist problems, the value of another simple clinical test in CSA was studied. Historically, the squeeze test, i.e. compression across the knuckles of MCP and metatarsophalangeal (MTP) joints, was used to assess presence of synovitis.⁴ In early arthritis a positive squeeze test was indeed shown to associate with presence of synovitis in MCP- and MTP-joints, and even with local MRI-detected inflammation.⁵ In the phase of CSA the squeeze test is considered a risk factor for development of RA, as it is also incorporated in the EULAR definition of arthralgia suspicious for progression to RA. It was therefore hypothesized that a positive squeeze test in CSA, in absence of clinical arthritis, might associate with presence of subclinical inflammation; this was investigated in **Chapter 3**. It was shown that >50% of CSA-patients had positive squeeze test in MCP- or MTP-joints, and that a positive test independently associated with local subclinical synovitis with an OR of 2. However, the sensitivity of the test was only 44%, indicating that subclinical synovitis is also often missed. In addition, a positive squeeze test in CSA was not independently associated with progression to IA. Nevertheless, despite its lack in predictive value, the squeeze test is a simple and quick test that can be used to obtain a first indication on presence of subclinical synovitis.

Considerations from Chapter 2 and 3

Chapter 2 and 3 describe two closely related tests that are part of the physical examination in clinical practice; difficulties making a fist and the squeeze test, both performed with different hand positioning. During development of the EULAR definition, experts indicated that both tests contributed to the recognition of CSA, and their independent contribution was confirmed in statistical analyses.³ In **Chapter 2 and 3** we aimed to increase our understanding of these tests, and discovered that they both associate with different inflammatory features; fist problems with tenosynovitis and the squeeze test with synovitis. This confirms the notion that both tests are in fact different and correlate with different types of subclinical inflammatory features. This suggests that both tests could be of value in the recognition of CSA in clinical practice.

Together, assessment of fist problems and the squeeze test help provide a first impression of underlying subclinical inflammation. Importantly, both studies were performed in a population of CSA-patients. The results can therefore not be generalized to other populations without further research. Primary care is a population where both tests might be of value in establishing a first risk assessment, since general physicians acknowledge the importance but also the difficulty of differentiating inflammatory diseases from other musculoskeletal problems.^{6,7} However, the predictive value of a test is dependent on the prevalence of disease in a population, i.e. the pre-test probability. The incidence of RA, and presumably also the prevalence of subclinical inflammation in primary care is low. Therefore it is likely that the predictive value of these tests is also lower in primary care than in CSA. Nevertheless, since test characteristics are unaffected by prevalence, the sensitivity and specificity will remain the same in primary care.

Summary Chapter 4

When presence of subclinical inflammation in at-risk populations is confirmed with imaging, treatment is sometimes considered, even in absence of clinical arthritis. Studies have shown that an increasing number of rheumatologists consider or initiate DMARD-treatment in patients with autoantibody-positive arthralgia,⁸ their choices guided by US findings and presence of subclinical inflammation.⁹ Indeed, subclinical inflammation can precede development of IA, but subclinical inflammation and symptoms also often spontaneously resolve.¹⁰ In **Chapter 4** we therefore addressed the value of subclinical synovitis, its potential as starting point for DMARD-treatment and its potential for overtreatment. We studied three arthralgia cohorts in which the presence of subclinical synovitis was determined at baseline by either US or MRI. All patients were followed for one year for development of IA, during which DMARDtreatment (including corticosteroids) was not allowed. In anti-citrullinated protein antibody (ACPA) positive patients with subclinical synovitis 50-68% of patients did not develop IA, in ACPA-negative patients 66-89% did not progress. Even in patients with additionally ≥ 6 points on the 2010 EULAR classification criteria for RA, false positive rates remained considerable (≥37%). Results also remained similar when more stringent definitions of subclinical synovitis were used in sensitivity analyses. These findings indicate that DMARD-treatment in arthralgia-patients with subclinical synovitis would lead to considerable overtreatment.

Considerations from Chapter 4

In **Chapter 4** both MRI and US were used. It has been shown that MRI has a higher sensitivity than US.¹¹ Nevertheless, false-positive rates in all three cohorts were high, and the use of different imaging modalities with varying sensitivity has therefore unlikely influenced conclusions from this study. False positive rates might be further decreased by additional evaluation of other inflammatory features, e.g. tenosynovitis and/or BME. The latter can only be visualized by MRI, and the predictive value independent from tenosynovitis is limited.² Tenosynovitis is detectable by both imaging modalities, and the predictive value is highest of all inflammatory features.^{2,12} It would therefore be valuable to repeat this study with subclinical tenosynovitis as

potential starting point for DMARD-treatment.

Summary Chapter 5

Subclinical inflammation as measured in Chapter 2, 3 and 4, even combined with ACPA-status, is insufficient for reliable identification of patients that progress to IA. Other imaging factors potentially increase the prognostic value of subclinical inflammation. Bone erosions are a hallmark of RA, and even RA-specific MRIdetected erosions have been established; these erosions were present in patients with early RA, but not in patients with other arthritides.¹³ MRI is sensitive in detection of bone erosions, even in symptom-free persons¹⁴ and in patients with CSA small MRI-detected erosions are detectable. In **Chapter 5** we investigated MRI-detected erosions in the phase of CSA. We determined the predictive value of MRI-detected erosions for development of IA, and evaluated whether the prognostic value of MRI-detected subclinical inflammation could be improved by evaluation of MRIdetected erosions. Any MRI-erosion, defined as erosions that were present in <5% of symptom-free persons in the same bone and age category, was present in 20% of CSApatients. Presence of these erosions was not associated with IA-development. Erosion characteristics previously reported as specific for RA (grade ≥ 2 erosions, erosions in MTP5 and erosions in MTP1 in persons aged <40) were rarely seen in CSA-patients, and their presence was not associated with IA-development. When MRI-detected erosions were considered in addition to MRI-detected subclinical inflammation, the area under the curve (AUC) did not improve, and the prognostic accuracy decreased as shown by a net reclassification index (NRI) of -5.8; adding data on MRI-erosions resulted in a high number of false-positive predictions. Since erosions mainly occur early in ACPA-positive disease,¹⁵⁻¹⁷ MRI-erosions were also evaluated in ACPA-positive and ACPA-negative CSA-patients separately. In neither subset MRI-detected erosions were predictive for development of IA. However, the median erosion score in ACPApositive patients was significantly higher than in ACPA-negative patients. Notably, this difference was only seen in patients with subclinical inflammation; ACPA-positive patients without subclinical inflammation did not have a higher erosion score than ACPA-negative patients without subclinical inflammation.

Considerations from Chapter 5

Findings in **Chapter 5** are supported by previous studies in undifferentiated arthritis (UA). The prevalence of RA-specific erosions was similar between UA- and CSA-patients,¹⁸ which is in line with the finding that erosion-scores in CSA-patients did not increase during progression to IA.¹⁹ Additionally, in UA-patients erosions also lacked predictive value for development of RA.¹⁸ Nevertheless, more erosions were present in CSA compared to symptom-free controls. Even though the exact mechanism is unclear, the erosions in CSA might reflect previous subclinical inflammation that, due

to lack of other stimuli, spontaneously resolved. Intriguingly, a significantly higher number of erosions was seen in ACPA-positive CSA-patients, though only when subclinical inflammation was present. This supports the finding that development of erosions in ACPA-positive CSA is mediated by subclinical inflammation, as shown previously.¹⁵ Potentially also a direct effect of ACPA on osteoclastogenesis and bone resorption exists, as indicated by a study from Harre et al.²⁰ However, since the number of erosions was not increased in CSA-patients with only ACPA (i.e. without subclinical inflammation), our data could not support this finding, Further research unravelling the mechanism between ACPA and development of erosions is needed. Because even though small MRI-detected erosions cannot be used as predictor for imminent RA, this study indicated that erosions already occur in the phase of CSA. Knowledge on the interplay between pro-inflammatory factors and ACPA in this process might direct future prevention of bone damage early in the disease.

Overall considerations from Part I

All longitudinal studies in **Part I** of this thesis were potentially influenced by the Treat Earlier trial, a randomized controlled trial (RCT) carried out in CSA-patients.²¹ From April 2015 until September 2019 CSA-patients with MRI-detected subclinical inflammation could be included in this trial, in which the effect of Methotrexate (MTX) was studied. Baseline characteristics of included patients could be used for cross-sectional analyses within the CSA-cohort. However, since the outcome of the CSA-cohort (development of IA) was potentially influenced by the 50% chance of DMARD-treatment in patients participating in this trial, these patients had to be excluded from longitudinal analyses. Excluding part of the patients with subclinical inflammation, a known risk factor for development of RA, might have influenced the associations of investigated predictors. Robustness of our findings has been investigated by stratification for subclinical inflammation, evaluation of only patients included in the CSA-cohort before the start of the trial and/or validation in other at-risk cohorts; these analyses consistently indicated similar results. It is therefore considered unlikely that the exclusion of part of the patients in longitudinal analyses caused incorrect results.

The predictors investigated in this part of the thesis were insufficient in establishing a CSA-population that would progress to clinical arthritis with certainty. MRIdetected erosions were not helpful and even increased the number of false-positive predictions. Subclinical synovitis was present in a high number of CSA-patients that did not progress to IA, and the squeeze test as proxy of subclinical synovitis did not yield predictive value. Likewise, despite tenosynovitis having the highest predictive value of all investigated features, not all CSA-patients with subclinical tenosynovitis, or with difficulties making a fist as proxy thereof, progressed to IA. It seems necessary to combine multiple predictors to obtain high predictive values. Indeed, prediction rules including known clinical and immunological predictors,²² combinations of MRI features,¹² or even both²³ have been shown to increase the AUC and positive predictive values (PPVs). Newly discovered imaging features might further improve these models. The discovery of other juxta-articular features as intermetatarsal bursitis (IMB), which has been associated with early RA²⁴ and conferred risk for arthritis development in CSA,²⁵ might be used for this purpose.

In addition, imaging techniques continuously improve, which may further enhance implementation of imaging in clinical practice. Shorter MRI scanning protocols without contrast-enhancement using modified Dixon sequences provide similar results as MRI sequences used in this thesis, though less costly and more patientfriendly.²⁶ Additionally, a decrease in workload for physicians and researchers, as well as an increase in faster and more consistent scoring methods is pursued by investigating possibilities of automated scoring methods with artificial intelligence (AI).

Nevertheless, despite improvements of imaging methods and discovery of new predictive features, advanced imaging modalities are not always (directly) available. It remains important to keep investigating associations of clinical features with (new) predictive imaging features, such as the squeeze test and difficulties making a fist which proved useful in this thesis. Their value should be investigated in future prediction models, which can improve applicability of prediction models in clinical practice and optimize a first risk assessment. Since presence of subclinical extensor peritendinitis of MCP-joints has been shown to have a high predictive value,¹² determining a clinical proxy for this feature might prove valuable in clinical practice.

Autoantibodies are among the first factors thought of in prediction of RA. ACPA and RF are among the strongest predictors for development of RA, and also play an important role in the 2010 EULAR classification criteria for RA.²⁷ Nevertheless, in up to 50% of RA-patients autoantibodies are absent. Even though the long-term outcomes of autoantibody-negative patients are less severe, clinical presentation and functional limitations are just as severe as in autoantibody-positive patients.²⁸ This underlines the importance of identifying not only early autoantibody-positive RA, but also early autoantibody-negative RA. At the moment most at-risk populations are composed of (arthralgia-)subjects with ACPA and/or RF, or relatives of RA-patients (e.g. first-degree relatives (FDR)). The CSA-cohort consists of a unique at-risk population, its inclusion based on clinical presentation rather than autoantibody-negative RA, large cohorts for validation are lacking. Even though the recognition of autoantibody-negative RA is more complicated by the absence of known immunological risk factors, it deserves

more attention in pre-clinical research.

Even when existing prediction rules are improved, it is likely that part of at-risk patients will still not progress to arthritis, despite presentation with one or more known predictors. This complicates early preventive treatment, as it will partly result in overtreatment. Overtreatment is not without consequences; DMARD-treatment is expensive and may cause side effects. However, as functional limitations in patients with CSA that later progress to IA are already as severe as at the moment of arthritis development,²⁹ treatment may significantly benefit some at-risk patients. To what extent overtreatment is acceptable is an important topic of discussion. Apart from the risk for overtreatment, preventive treatment also complicates new studies in the prearthritis phase. The natural disease course that is presently studied in observational cohorts is then influenced by potential treatment responses in part of the patients. Moreover, ethical concerns may arise when initiating clinical trials investigating new preventive treatment in which part of the patients would receive placebo treatment. Lastly, the effectiveness of treatment in prevention of RA is not clear yet,³⁰⁻³² and currently still investigated in several trials.^{21,33-35} It is therefore recommended to await clinical trial results and optimize prediction rules before starting DMARD-treatment in populations where clinical arthritis is not yet present.

Part II - Pathogenesis of rheumatoid arthritis

Adequate identification of patients at risk for development of RA can be improved by understanding the disease pathogenesis. Knowledge on disease processes and timing of contributing factors during initiation and progression of the disease help to better target treatment in pre-clinical stages, ultimately preventing RA.

Summary Chapter 6

Autoantibody development and response maturation, defined as an increase in number of autoantibodies and autoantibody levels, precede the development of RA.³⁶⁻³⁸ It was unknown whether autoantibody-response maturation occurred in the phase of CSA, or whether the response was already fully matured at the onset of symptoms. In **Chapter 6** we therefore evaluated the presence and levels of autoantibodies in CSA at two timepoints; in patients that progressed to IA samples were taken at baseline and at the moment of IA-development, in patients that did not progress samples were taken at baseline and after two years. If maturation of the autoantibody-response played a role in disease progression, maturation was expected to be only present in CSA-patients that developed IA. We analyzed three autoantibodies (ACPA, anti-carbamylated protein antibodies (anti-CarP) and anti-

acetylated protein antibodies (AAPA)) in three different isotypes (IgM, IgG and IgA). Patients without any autoantibody at baseline rarely seroconverted to positive during follow-up. In patients with \geq 1 autoantibody (out of nine) at baseline the median number of autoantibodies was 1, and an increase in number of autoantibodies was infrequent. Autoantibody levels did not significantly change over time. These findings were similar between patients that progressed to IA and patients that did not progress. We therefore concluded that autoantibody-response maturation was not responsible for the final hit in development of clinical arthritis. However, when the outcome RA was used, i.e. fulfilment of the 1987 and/or 2010 ACR/EULAR criteria^{27,39} at the moment of IA-development, patients who did not progress showed a decrease in median number of autoantibody-response were lacking in these patients.

Considerations from Chapter 6

Due to laboratory capacity autoantibodies at two time-points could only be measured in part of CSA-patients. This predominantly affected the selection of non-progressing CSA-patients that tested negative for RF and/or ACPA during routine laboratory measurements at baseline; from this group a random sample was studied. Preferably the entire CSA-population would have been studied. However, since baseline characteristics of included and excluded patients were similar, it is unlikely that replication of this study in the entire CSA-population would yield different results. Ideally, population-based studies are performed to confirm that autoantibodyresponse maturation occurs predominantly in the asymptomatic, and not in the symptomatic phase of RA-development.

Summary Chapter 7

In **Chapter 7** we focused on the role and timing of the two most prominent genetic and environmental risk factors for development of RA; HLA-SE and smoking. Their association with RA is widely acknowledged, though it is unknown at which disease stage they exert their effect. To investigate a potential role of HLA-SE and smoking in autoantibody-development we studied literature on associations of HLA-SE and smoking with presence of ACPA in the asymptomatic population. Meta-analyses revealed that smoking, but not HLA-SE, was associated with ACPApositivity in asymptomatic individuals. At presentation with symptoms (CSA-onset) both HLA-SE and smoking associated with presence of ACPA. Though previous studies showed gene-environment interactions for development of ACPA,^{40,41} in CSA the association of smoking with ACPA was not dependent on presence of HLA-SE, and no significant interaction between HLA-SE and smoking was found. Likewise, previous findings in RA indicating that smoking was not associated with ACPA, but with RF or autoantibodies in general,^{42,43} could not be replicated in meta-analyses of asymptomatic populations or in CSA-patients. During follow-up HLA-SE associated with progression to IA in the total CSA-population, as well as in the ACPA-positive subset as indicated by meta-analyses in three arthralgia-cohorts. Smoking was not associated with IA-development, not in the total CSA-population, after ACPA-stratification or in meta-analyses with other cohorts. Together, results from this study imply that smoking is involved in autoantibody-development, and possibly symptom-development, but not with further progression to IA. The HLA-SE is not involved in autoantibody-development, and symptom-development, and associates with IA-development.

Considerations from Chapter 7

Importantly, conclusions from **Chapter 7** are partly based on results from crosssectional data. Associations of smoking and HLA-SE with autoantibody-development were investigated in cross-sectional studies of asymptomatic populations. However, we believe that autoantibody-development is the first step towards development of autoantibody-positive RA. Therefore it is likely that these cross-sectional data accurately reflect the roles of smoking and HLA-SE in autoantibody-development. Associations of smoking and HLA-SE with autoantibody-response maturation and symptom-development have been based on data obtained at CSA-onset. Though different studies support findings of smoking having a role in progression to CSA,⁴⁴ and HLA-SE being involved in autoantibody-response maturation,⁴⁵ longitudinal data are needed to confirm these findings. Ideally, population-based studies following individuals from an asymptomatic stage towards development of RA are performed. Such studies are complicated by the low prevalence of RA in the general population; an excessive number of subjects is needed to detect even a few that eventually develop RA.

In **Chapter 7** we could not find associations between smoking, HLA-SE and autoantibodies other than ACPA, i.e. RF, AAPA and anti-CarP. Even though our goal was to determine when certain risk factors exerted their effected, we also explored the predictive value of AAPA and anti-CarP. Thus far conflicting findings on predictive and additional value of anti-CarP were reported, and AAPA was never studied in arthralgia-patients. With data from two arthralgia-cohorts we showed that only AAPA, but not anti-CarP, associated with development of IA independently from ACPA and RF. Further research is necessary to validate our findings.

Overall considerations from Part II

Until large population-based studies are performed, we can only speculate about the biological mechanisms behind our findings in **Chapter 6 and 7**. Since no association was found between HLA-SE and autoantibodies in the asymptomatic stage, it seems

likely that smoking, rather than HLA-SE, stimulates the initial break of tolerance to citrullinated antigens. Autoantibody-response maturation is likely stimulated by HLA-SE, which is suggested by the tendency of HLA-SE to associate with higher ACPA-levels. An increase in autoantibody levels or number of autoantibodies and isotypes occurs before symptom-development. After symptom-onset the autoantibody-response, as measured in **Chapter 6**, generally remains unchanged and does not provide a final hit towards development of IA. Other forms of autoantibody maturation than measured in **Chapter 6**, e.g. changes in cross-reactivity, affinity maturation, epitope spreading and glycosylation profile, could occur in the symptomatic phase. It is tempting to hypothesize that these forms of maturation might be involved in the final hit towards development of clinical arthritis. However, since these autoantibody characteristics associate with autoantibody levels, it seems more likely that the final hit is influenced by factors other than the autoantibody response, e.g. by other processes of the adaptive immune system or other (yet) unknown factors unrelated to the autoantibody response.

Our findings suggest that the HLA-SE might play a role in the final hit, since it significantly associated with IA-development in ACPA-positive patients in **Chapter 7**. An HLA-SE restricted T-cell response potentially stimulates the already existing ACPA-response. ACPA-IgG variable-domain glycosylation has indeed been shown to increase towards symptom-onset and significantly associates with HLA-SE.⁴⁶ While the exact mechanism remains to be elucidated, influences of HLA-SE on the ACPA-response could explain why ACPA-positive patients with HLA-SE more often develop RA than ACPA-positive patients without HLA-SE. Nevertheless, since also RA-patients without HLA-SE and/or autoantibodies present with similar clinical manifestations, other triggers remain to be elucidated.

As previously addressed in **Part I** of this thesis, a large part of RA-patients is autoantibody-negative. Apart from improving prediction in this RA-subset, knowledge on disease pathogenesis of autoantibody-negative disease needs to be enhanced. In **Chapter 7** initial analyses suggested that HLA-SE was somewhat predictive for development of ACPA-negative IA. However, this could not be replicated with the outcome RA. Since most at-risk cohorts only include autoantibody-positive patients we were not able to further explore these findings. Nevertheless, previous research suggested that HLA-SE is also involved in ACPA-negative RA, though with a smaller effect size.⁴⁷ This suggests that an effect of HLA-SE with RA-development might be found when studying a larger group of autoantibody-negative at-risk patients.

Future perspectives

Early detection and knowledge on disease pathogenesis of RA have tremendously improved over recent years, and first steps towards prevention have been taken. Nevertheless, further research is necessary to improve risk stratification and understanding of disease development.

The following points might be topic of future studies:

- A search for new clinical tests that associate with predictive imaging features, e.g. for MCP extensor peritendinitis which has shown high predictive value in CSA.
- The value of difficulties making a fist and the squeeze test may be tested in other populations, i.e. primary care, where they might prove valuable in determining presence of subclinical inflammation or risk for inflammatory disease.
- Further optimization of prediction rules, in which the value of newly discovered imaging features (e.g. juxta-articular inflammation as inter-metatarsal bursitis) or biomarkers (e.g. AAPA and/or ACPA-IgG glycosylation) need to be determined. With the prospect of preventive treatment in the future, the risk of overtreatment must be evaluated during development of new prediction rules.
- Findings on autoantibody-response maturation and timing of effects of HLA-SE and smoking as described in this thesis are ultimately confirmed in longitudinal population-based studies.
- Further research on disease pathogenesis to establish which factors are involved in the final hit towards development of RA. Factors may concern other forms of autoantibody maturation than addressed in this thesis, other processes of the adaptive immune system or even (yet) unknown factors entirely unrelated to the autoantibody response. Knowledge on timing of contributing factors may help to better target treatment in pre-clinical stages.
- Development of autoantibody-negative at-risk cohorts. While the CSA-cohort is useful in identification of predictors for autoantibody-negative RA, large cohorts for validation are lacking. Apart from identification of predictors, additional autoantibody-negative cohorts can be used to further elucidate disease processes in development of autoantibody-negative RA.

Final conclusions

In this thesis we have investigated CSA-patients and reported on the value of several clinical, imaging and immunological factors for prediction of RA. We also added to knowledge on disease pathogenesis and timing of disease processes by investigating immunological, genetic and environmental factors contributing to RA-development

in both asymptomatic and symptomatic disease phases. Although progress has been made, with the current knowledge and risk stratification it is not yet recommended to start preventive DMARD-treatment outside research settings in the absence of clinically apparent inflammatory arthritis. To prevent overtreatment, it is necessary to refrain from DMARD-treatment until adequate risk assessment is established, and clinical trials investigating the effect of preventive treatment have proven its value. Until such times, research should focus on the natural disease course of RAdevelopment, further optimizing prediction and knowledge on disease mechanisms. In time, this could ultimately lead to prevention of RA in a high risk population, with the right treatment, at the right time.

References

- 1. van Beers-Tas MH, Blanken AB, Nielen MMJ et al. The value of joint ultrasonography in predicting arthritis in seropositive patients with arthralgia: a prospective cohort study. Arthritis Res Ther. 2018;20:279.
- van Steenbergen HW, Mangnus L, Reijnierse M et al. Clinical factors, anticitrullinated peptide antibodies and MRI-detected subclinical inflammation in relation to progression from clinically suspect arthralgia to arthritis. Ann Rheum Dis. 2016;75:1824-30.
- 3. van Steenbergen HW, Aletaha D, Beaartvan de Voorde LJ et al. EULAR definition of arthralgia suspicious for progression to rheumatoid arthritis. Ann Rheum Dis. 2017;76:491-6.
- 4. Firestein GS, Budd R, Gabriel SE et al. Kelley's Textbook of Rheumatology: Elsevier Health Sciences; 2012.
- van den Bosch WB, Mangnus L, Reijnierse M et al. The diagnostic accuracy of the squeeze test to identify arthritis: a cross-sectional cohort study. Ann Rheum Dis. 2015;74:1886-9.
- 6. Ten Brinck RM, van Dijk BT, van Steenbergen HW et al. Development and validation of a clinical rule for recognition of early inflammatory arthritis. BMJ open. 2019;8:e023552.
- 7. Meyfroidt S, Stevens J, De Lepeleire J et al. A general practice perspective on early rheumatoid arthritis management: A qualitative study from Flanders. Eur J Gen Pract. 2015;21:231-7.
- 8. van Boheemen L, Bolt JW, Ter Wee MM et al. Patients' and rheumatologists' perceptions on preventive intervention in rheumatoid arthritis and axial spondyloarthritis. Arthritis Res Ther. 2020;22:217.
- 9. Mankia K, Briggs C, Emery P. How Are Rheumatologists Managing Anticyclic Citrullinated Peptide Antibodies-positive Patients Who Do Not Have Arthritis? J Rheumatol. 2020;47:305-6.
- 10. Ten Brinck RM, Boeters DM, van Steenbergen HW et al. Improvement of symptoms in clinically suspect arthralgia and resolution

of subclinical joint inflammation: a longitudinal study in patients that did not progress to clinical arthritis. Arthritis Res Ther. 2020;22:11.

- 11. Ohrndorf S, Boer AC, Boeters DM et al. Do musculoskeletal ultrasound and magnetic resonance imaging identify synovitis and tenosynovitis at the same joints and tendons? A comparative study in early inflammatory arthritis and clinically suspect arthralgia. Arthritis Res Ther. 2019;21:59.
- 12. Matthijssen XME, Wouters F, Boeters DM et al. A search to the target tissue in which RAspecific inflammation starts: a detailed MRI study to improve identification of RA-specific features in the phase of clinically suspect arthralgia. Arthritis Res Ther. 2019;21:249.
- 13. Boeters DM, Nieuwenhuis WP, van Steenbergen HW et al. Are MRI-detected erosions specific for RA? A large explorative cross-sectional study. Ann Rheum Dis. 2018;77:861-8.
- 14. Mangnus L, van Steenbergen HW, Reijnierse M et al. Magnetic Resonance Imaging-Detected Features of Inflammation and Erosions in Symptom-Free Persons From the General Population. Arthritis Rheumatol. 2016;68:2593-602.
- 15. Ten Brinck RM, Toes REM, van der Helmvan Mil AHM. Inflammation functions as a key mediator in the link between ACPA and erosion development: an association study in Clinically Suspect Arthralgia. Arthritis Res Ther. 2018;20:89.
- 16. Kleyer A, Finzel S, Rech J et al. Bone loss before the clinical onset of rheumatoid arthritis in subjects with anticitrullinated protein antibodies. Ann Rheum Dis. 2014;73:854-60.
- 17. Berglin E, Johansson T, Sundin U et al. Radiological outcome in rheumatoid arthritis is predicted by presence of antibodies against cyclic citrullinated peptide before and at disease onset, and by IgA-RF at disease onset. Ann Rheum Dis. 2006;65:453-8.
- Boeters DM, Boer AC, van der Helm-van Mil AHM. Evaluation of the predictive accuracy of MRI-detected erosions in hand and foot joints in patients with undifferentiated

arthritis. Ann Rheum Dis. 2019;78:144-6.

- Ten Brinck RM, van Steenbergen HW, van der Helm-van Mil AHM. Sequence of joint tissue inflammation during rheumatoid arthritis development. Arthritis Res Ther. 2018;20:260.
- Harre U, Georgess D, Bang H et al. Induction of osteoclastogenesis and bone loss by human autoantibodies against citrullinated vimentin. J Clin Invest. 2012;122:1791-802.
- 21. Niemantsverdriet E, Dakkak YJ, Burgers LE et al. TREAT Early Arthralgia to Reverse or Limit Impending Exacerbation to Rheumatoid arthritis (TREAT EARLIER): a randomized, double-blind, placebo-controlled clinical trial protocol. Trials. 2020;21:862.
- 22. van de Stadt LA, Witte BI, Bos WH et al. A prediction rule for the development of arthritis in seropositive arthralgia patients. Ann Rheum Dis. 2013;72:1920-6.
- 23. Matthijssen X, Van Dijk B, Wouters F et al. FRI0542 OBTAINING HIGH POSITIVE PREDICTIVE VALUES FOR THE DEVELOPMENT OF CLINICALLY APPARENT ARTHRITIS IN PATIENTS PRESENTING WITH CLINICALLY SUSPECT ARTHRALGIA; IS IT FEASIBLE? Annals of the Rheumatic Diseases. 2020;79:872-.
- 24. Dakkak YJ, Niemantsverdriet E, van der Helm-van Mil AHM et al. Increased frequency of intermetatarsal and submetatarsal bursitis in early rheumatoid arthritis: a large case-controlled MRI study. Arthritis Res Ther. 2020;22:277.
- 25. van Dijk BT, Wouters F, van Mulligen E et al. During development of rheumatoid arthritis, intermetatarsal bursitis may occur before clinical joint swelling: a large imaging study in patients with clinically suspect arthralgia. Rheumatology (Oxford). 2021.
- 26. Niemantsverdriet E, Verstappen M, Wouters F et al. POS0257 TOWARDS A SIMPLIFIED FLUID-SENSITIVE MRI-PROTOCOL IN SMALL JOINTS OF THE HAND IN EARLY ARTHRITIS PATIENTS: RELIABILITY BETWEEN MDIXON AND REGULAR FSE FAT SATURATION MRI-SEQUENCES. Annals of the Rheumatic Diseases. 2021;80:351-.
- 27. Aletaha D, Neogi T, Silman AJ et al. 2010 Rheumatoid arthritis classification criteria: an American College of Rheumatology/

European League Against Rheumatism collaborative initiative. Arthritis Rheum. 2010;62:2569-81.

- Boer AC, Boonen A, van der Helm van Mil AHM. Is Anti-Citrullinated Protein Antibody-Positive Rheumatoid Arthritis Still a More Severe Disease Than Anti-Citrullinated Protein Antibody-Negative Rheumatoid Arthritis? A Longitudinal Cohort Study in Rheumatoid Arthritis Patients Diagnosed From 2000 Onward. Arthritis Care Res (Hoboken). 2018;70:987-96.
- 29. Ten Brinck RM, van Steenbergen HW, Mangnus L et al. Functional limitations in the phase of clinically suspect arthralgia are as serious as in early clinical arthritis; a longitudinal study. RMD Open. 2017;3:e000419.
- Gerlag DM, Safy M, Maijer KI et al. Effects of B-cell directed therapy on the preclinical stage of rheumatoid arthritis: the PRAIRI study. Ann Rheum Dis. 2019;78:179-85.
- 31. Bos WH, Dijkmans BA, Boers M et al. Effect of dexamethasone on autoantibody levels and arthritis development in patients with arthralgia: a randomised trial. Ann Rheum Dis. 2010;69:571-4.
- 32. van Boheemen L, Turk S, Beers-Tas MV et al. Atorvastatin is unlikely to prevent rheumatoid arthritis in high risk individuals: results from the prematurely stopped STAtins to Prevent Rheumatoid Arthritis (STAPRA) trial. RMD Open. 2021;7.
- 33. Al-Laith M, Jasenecova M, Abraham S et al. Arthritis prevention in the pre-clinical phase of RA with abatacept (the APIPPRA study): a multi-centre, randomised, double-blind, parallel-group, placebo-controlled clinical trial protocol. Trials. 2019;20:429.
- 34. Strategy to Prevent the Onset of Clinically-Apparent Rheumatoid Arthritis (StopRA). NCT02603146. https://clinicaltrials.gov/ct2/ show/NCT02603146. Accessed 05-01-2022.
- 35. Abatacept Reversing Subclinical Inflammation as Measured by MRI in ACPA Positive Arthralgia (ARIAA). NCT02778906. https://clinicaltrials.gov/ct2/show/ NCT02778906. Accessed 05-01-2022.
- 36. Rantapaa-Dahlqvist S, de Jong BA, Berglin E et al. Antibodies against cyclic citrullinated peptide and IgA rheumatoid factor predict

the development of rheumatoid arthritis. Arthritis Rheum. 2003;48:2741-9.

- 37. Nielen MM, van Schaardenburg D, Reesink HW et al. Specific autoantibodies precede the symptoms of rheumatoid arthritis: a study of serial measurements in blood donors. Arthritis Rheum. 2004;50:380-6.
- 38. Gan RW, Trouw LA, Shi J et al. Anticarbamylated protein antibodies are present prior to rheumatoid arthritis and are associated with its future diagnosis. J Rheumatol. 2015;42:572-9.
- 39. Arnett FC, Edworthy SM, Bloch DA et al. The American Rheumatism Association 1987 revised criteria for the classification of rheumatoid arthritis. Arthritis Rheum. 1988;31:315-24.
- 40. Klareskog L, Stolt P, Lundberg K et al. A new model for an etiology of rheumatoid arthritis: smoking may trigger HLA-DR (shared epitope)-restricted immune reactions to autoantigens modified by citrullination. Arthritis Rheum. 2006;54:38-46.
- 41. Hedstrom AK, Ronnelid J, Klareskog L et al. Complex Relationships of Smoking, HLA-DRB1 Genes, and Serologic Profiles in Patients With Early Rheumatoid Arthritis: Update From a Swedish Population-Based Case-Control Study. Arthritis Rheumatol. 2019;71:1504-11.
- 42. van Wesemael TJ, Ajeganova S, Humphreys J et al. Smoking is associated with the concurrent presence of multiple autoantibodies in rheumatoid arthritis rather than with anti-citrullinated protein antibodies per se: a multicenter cohort study. Arthritis Res Ther. 2016;18:285.

- 43. Ishikawa Y, Ikari K, Hashimoto M et al. Shared epitope defines distinct associations of cigarette smoking with levels of anticitrullinated protein antibody and rheumatoid factor. Ann Rheum Dis. 2019;78:1480-7.
- 44. Westra J, Brouwer E, Raveling-Eelsing E et al. Arthritis autoantibodies in individuals without rheumatoid arthritis: follow-up data from a Dutch population-based cohort (Lifelines). Rheumatology (Oxford). 2020.
- 45. Bos WH, Ursum J, de Vries N et al. The role of the shared epitope in arthralgia with anticyclic citrullinated peptide antibodies (anti-CCP), and its effect on anti-CCP levels. Ann Rheum Dis. 2008;67:1347-50.
- 46. Kissel T, van Schie KA, Hafkenscheid L et al. On the presence of HLA-SE alleles and ACPA-IgG variable domain glycosylation in the phase preceding the development of rheumatoid arthritis. Ann Rheum Dis. 2019;78:1616-20.
- 47. Viatte S, Plant D, Bowes J et al. Genetic markers of rheumatoid arthritis susceptibility in anti-citrullinated peptide antibody negative patients. Ann Rheum Dis. 2012;71:1984-90.